Author's response to reviews

Title: Temporary epicardial cardiac resynchronisation versus conventional right ventricular pacing after cardiac surgery: study protocol for a randomised control trial

Authors:

Stuart J Russell (russellsj1@cf.ac.uk)
Christine Tan (christine.tan@wales.nhs.uk)
Peter O'Keefe (peter.o'keefe@wales.nhs.uk)
Saeed Ashraf (saeed.ashraf@wales.nhs.uk)
Afzal Zaidi (afzal.zaidi@wales.nhs.uk)
Alan G Fraser (fraserag@cf.ac.uk)
Zaheer R Yousef (zaheer.yousef@wales.nhs.uk)

Version: 7 Date: 19 February 2012

Author's response to reviews:

Dear Sir,

1. The 'group name' has been removed from the manuscript, we will just use are individual names for the paper.

2. The abstract on the website has been updated and is now the same as the final manuscript (version 9).

3. The references are in the correct sequence.

4. The endnote format has been altered and the term 'et al' has been removed from all references.

5. The paper has been rechecked for typographical errors.

Please let me know if I can provide any further information.

Yours sincerely,

Dr Stuart Russell.